Author's response to reviews

Title: Bilateral macular hemorrhage as a complication of drug induced anemia: A case report

Authors:

Rubens N Belfort (epm@belfort.med.br)
Bruno F Fernandes (bruno.mtl@gmail.com)
Ricardo Nose (ricnose@hotmail.com)
Andre Romano (andre@romano.med.br)
Jonathan Cools-Lartigue (jonathan.cools-lartigue@mail.mcgill.ca)
Eduardo V Navajas (edunavajas@gmail.com)
Garles M M Vieira (garles74@yahoo.com.br)
Renato Delascio Lopes (renato.lopes@duke.edu)
Rubens Belfort Jr. (clinbelf@uol.com.br)

Version: 2 Date: 14 August 2008

Author's response to reviews:

Editor

Montreal, August 14th, 2008

Kindly attention of the Editor,

Enclosed you will find our manuscript entitled “Bilateral macular hemorrhage as a complication of drug induced anemia”. We described in detail the case report of a rare complication in the treatment of ocular toxoplasmosis. We believe this is the first report of such complication in the literature. We will be pleased if you consider our paper for publication in the “Journal of Medical Case Reports”.

The authors warrant that this article is original. It has not been published before and is not intended to be published anywhere except in the “Journal of Medical Case Reports” in the event the work is accepted. The article has been prepared according to each and every one of your manuscript instructions.

As a main author, Rubens N Belfort ensures that had full access to all patients information and takes responsibility for the integrity of the data and the accuracy of the data as well as the decision to submit for publication. The article was seen and approved by all other authors. No financial support was received and the authors declare no conflict of interests.

We have addressed the modifications requested by e-mail on August 14 2008.

We hope this article will be viewed in a favorable light. Thank you very much in advance for your consideration and collaboration.
Sincerely yours,

Rubens N Belfort; Bruno F. Fernandes; André Romano; Eduardo V Navajas; Ricardo Nose; Jonathan Cools-Lartigue; Garles MM Vieira; Renato Delascio Lopes; Rubens Belfort Jr.